Author's response to reviews

Title: Primary extra-gastrointestinal stromal tissue tumor of the whole abdominal cavity, omentum, peritoneum and mesentery: a rare case report and review of the articles.

Authors:

Abdulmaged M Musaad (magid999@live.com)
Elsaggad E A (sagadgady@yahoo.com)
Abdulmunem A Abdo (munem2002@hotmail.com)
Ahmed M Elhassan (ahmedelhassan85@gmail.com)
Hiba Hassan (hibahassan@hotmail.com)
Nasreeldeen Adam (sulieman.nasreeldeen@facebook.com)
Mohamed Abdelazeem (mohamedazeem3@hotmail.com)
Nassir A Arabi (nassir_alhaboob@yahoo.com)

Version: 4 Date: 6 September 2013

Author's response to reviews: see over
This case adds to the medicine the uncommon presentation of EGIST and exhorts more study in this field.

It is not the first report of this kind of disease but a rare presentation, it significantly advance our understanding of this disease etiology and the difficulty that we can confront in the management, and it is an original case report of interest in surgery and surgical oncology, there is no competing interest.

To our knowledge we would like to publish this case report to share the idea of how we can get the best management regarding the difficulty in resection the tumour and reaching the cure level and the possibility of potent chemotherapy in the management to avoid recurrence.

Potential peer reviewer:

Prof Ake Andren Sandberg
ake.andren-sandberg@karolinska.se

prof nystrom
per-olof.nystrom@karolinska.se

dr wim ceelen
wim.ceelen@ugent.be